\documentclass[11pt,amssymbols]{article}

\begin{document}
\begin{center}
{\Large Math 6010  2005/06 Fall}
\end{center}
\vspace{.3 cm}
\begin{tabbing}
Textbook:  \= Linear Statistical Analysis\\
         \>by G.A.F. Seber and A.J. Lee, Second Ed. 2003\\
         \> (You could use the first edition if you have it)
\end{tabbing}
Instructor:  Lajos Horv\'{a}th (e-mail: horvath@math.utah.edu)\\
Office hour: Wednesday  1:00 pm --2:00 pm  JWB 222 or by appointment\\
\begin{tabbing}
Prereq: \= Math 5010 (Introduction to Probability Theory) \\
\> Math 5080--5090 (Introduction to Mathematical Statistics)\\
%\> Math 601 (Linear Models)\\
\end{tabbing}

\begin{tabbing}
Midterms:$\;$\=  September 30 (50 minutes)\\
        \>  October 28  (50 minutes)\\
        \> December 2    (50 minutes)
\end{tabbing}
Final:  None
%
\begin{tabbing}
Project deadlines:$\;$\=  September 23 \\
        \>  October 21\\
        \> November 23
\end{tabbing}

\begin{tabbing}
Computation of the final grade: $\;$ %\= Projects: \quad \quad  \=30\% \\
\= Midterm  1: \quad \quad  \= 15\%\\
\> Midterm  2: \> 15\%\\
\> Midterm  3: \> 15\%\\
\> Project  1: \> 15\%\\
\> Project  2: \> 15\%\\
\> Project  3: \> 15\%\\
\> Presentation \> 10\%\\
\> Participation\> 10\%\\
%\> Final :  \> 30 \%
\end{tabbing}

\begin{tabbing}
A    \quad    \= : 91--100   \quad \quad      \= C  \quad     \=: 60--64\\
A--   \quad   \>: 85--90             \>C--      \> : 55--59\\
B+   \quad    \>: 80--84            \>D+       \>: 50--54\\
B    \quad    \>: 75--79              \>D       \>: 40--49\\
B--  \quad    \>: 70--74              \>D--       \>: 25--39\\
C+  \quad    \>:  65--69             \>E         \>:  0--24 \\
\end{tabbing}

\medskip
All projects must be typed using tex or latex. It must be written as a report and it should be readable i.e. the students in this class should understand what you want to say. Graphs and tables should be part of the tex. Graphs produced by SAS are not acceptable. \\
You will present one of the project in the class. It should last for 10--15 min. \\
The first month will be spent on linear algebra and this will be the topic of the first midterm.\\
You should be working on your project alone. It is OK to discuss technical problems, like how to use softwares or how to type in latex with other students or with the computer guys of the mathematics department. However, the basic idea of the project should be yours.



\end{document}
